Author's response to reviews

Title: Epidemiology of frequent attenders: a 3-year historic cohort study comparing attendance, morbidity and prescriptions of one-year and persistent frequent attenders

Authors:

Frans Th.M. Smits (fsmits@gazocentra.nl)
Henk J. Brouwer (h.j.brouwer@amc.uva.nl)
Gerben ter Riet (g.terriet@amc.uva.nl)
Henk H.C. van Weert (h.c.vanweert@amc.uva.nl)

Version: 6 Date: 14 January 2009

Author's response to reviews:

Sorry. Two small corrections.
- page 4: ref 12 was mentioned twice.
- table 1: the superscript was not correct.